Author's response to reviews

Title: Renal cell carcinoma presenting with intramyocardial metastases

Authors:

Anna M Czarnecka (anna.czarnecka@gmail.com)
Pawel Sobczuk (paweljp@interia.eu)
Fei Lian (fei.lian@emory.edu)
Cezary Szczylik (cszczylik@wp.pl)

Version: 2 Date: 5 April 2014

Author's response to reviews:

1. Email Address of all authors was listed in the title page of manuscript.
2. Line numbers were included in manuscript.
3. Manuscript was set into double spacing format.